Author's response to reviews

Title: Histological analysis of the effects of a static magnetic field on bone healing process in rat femurs

Authors:

Edela Puricelli (epuricelli@uol.com.br)

Version: 2 Date: 29 January 2007

Author's response to reviews: see over
18 January 2007

To
Priv.-Doz. Dr. med. dent. Thomas Stamm
Editor-in-Chief, HEAD & FACE MEDICINE

Dear Sir

I would like to submit the manuscript “A new technique for mandibular osteotomy” to your attention for publication in the Head & Face Medicine. The work is unpublished, and there is no significant overlap between the submitted manuscript and any other paper under consideration or in press elsewhere. The manuscript does not include subjects about whom a previous report has been published.

In thanking you for your attention, I look forward to hearing from you.

Yours sincerely

Prof Dr Edela Puricelli
School of Dentistry, Federal University of Rio Grande do Sul
Porto Alegre, RS, Brazil